Author's response to reviews

Title: The importance of central corneal thickness measurements and decision making in general ophthalmology clinics: a masked observational study

Authors:

Ashish A Patwardhan (apatwa@hotmail.com)
Mohammad I Khan (drmikhan@gmail.com)
Susan P Mollan (soozmollan@doctors.org.uk)
Paul M Haigh (paul.haigh@rsh.nhs.uk)

Version: 5 Date: 22 November 2007

Author's response to reviews: see over
To
The Editor
BMC Ophthalmology
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB
UK

Dear Sir/Madam

I have attached the revised manuscript incorporating changes suggested by the reviewer. We accept reviewer Scott Fraser’s comment about the terminology and we have changed terminology from LTG (Low Tension Glaucoma) to NPG (Normal Pressure Glaucoma). I hope the revised manuscript will meet the standard for publication in your journal.

Yours sincerely

Ashish Patwardhan